Author's response to reviews

**Title:** RIDOM: Comprehensive and public sequence database for identification of Mycobacterium species

**Authors:**

Dr Dag Harmsen (dharmsen@uni-muenster.de)
Stefan Dostal (sdostal@web.de)
Dr Andreas Roth (mikromau@zedat.fu-berlin.de)
Stefan Niemann (sniemann@fz-borstel.de)
Jorg Rothganger (jrothgaenger@ridom.de)
Michael Sammeth (micha@sammeth.net)
Prof Jurgen Albert (mfrosch@hygiene.uni-wuerzburg.de)
Prof Matthias Frosch (mfrosch@hygiene.uni-wuerzburg.de)
Elvira Richter (erichter@fz-borstel.de)

**Version:** 3  **Date:** 7 Nov 2003

Submission of a revised manuscript for publication in the "BMC Infectious Disease" Journal (MS: 1092583128214161)

Dear Editors,

please find enclosed a complete copy of a revised manuscript titled "RIDOM: Comprehensive and public sequence database for identification of Mycobacterium species" by Dag Harmsen, Stefan Dostal, Andreas Roth, Stefan Niemann, Jorg Rothganger, Michael Sammeth, Jurgen Albert, Matthias Frosch, and Elvira Richter. As suggested in your email dated from 4th November we have now attached our long tables 1 and 2 as additional files to the manuscript. Many thanks in advance for all of your efforts!

Yours sincerely,

Dr. Dag Harmsen
Email: dharmsen@uni-muenster.de, Phone: +49-251/8355348, and Fax: +49-251/8355341